Author’s response to reviews

Title: Contraceptive use among lactating women in Ganta-Afeshum District, Eastern Tigray, Northern Ethiopia, 2015: A cross sectional study

Authors:

Alem Gebremariam (alemg25@gmail.com)
Hadush Gebremariam (hadushgm@gmail.com)

Version: 2 Date: 23 Nov 2017

Author’s response to reviews:

Alem Gebremariam

E-mail: alemg25@gmail.com
Phone number- +251-910352915

Adigrat, Ethiopia

Title: Contraceptive use among lactating women in Ganta-Afeshum District, Eastern Tigray, Northern Ethiopia, 2015: A cross sectional study

Dear Ines Lopes de Sousa,

I am delighted that the manuscript “Contraceptive use among lactating women in Ganta-Afeshum District, Eastern Tigray, Northern Ethiopia, 2015: A cross sectional study” (PRCH-D-17-00271R1) has been assessed by your reviewers and deemed potentially acceptable for publication in BMC Pregnancy and Childbirth. We have read the manuscript and designed according to the authors’ guideline in the BMC Pregnancy and childbirth manuscript submission page. We have proofread it and made some topographic corrections. Moreover, Mr. Dean Withroder, from America, University of Utah School of Medicine, also read and edited the manuscript.
We also enclose the revised manuscript.

Kindest regards,

Yours faithfully,

Alem Gebremariam (BSc, MPH)

On behalf of my co-authors